Author's response to reviews

Title: A Controlled Trial of Automated Classification of Negation from Clinical Notes

Authors:

Peter L Elkin (elkin.peter@mayo.edu)
Steven H Brown (Steven.Brown@med.va.gov)
Brent A Bauer (bauer.brent@mayo.edu)
Casey S Husser (husser.casey@mayo.edu)
Bill Carruth (bcarruth@jhmi.edu)
Larry R Bergstrom (bergstrom.larry@mayo.edu)
Dietlind L Wahner-Roedler (wahnerroedler.dietlind@mayo.edu)

Version: 6 Date: 25 March 2005

Author's response to reviews: see over
March 25, 2005

Iratxe Puebla, Assistant Editor
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Dear Asst. Editor:

I have carried out additional formatting changes to resubmit my manuscript according to the instructions requested?

**Title** – I used sentence case in the title provided in the submission system (i.e. removed all unnecessary capitals).

**Country** – I included the country in the affiliation details.

**Corresponding author** – I added corresponding author on the title page with email address given

**Abstract** – I updated the abstract details in the submission system.

**Manuscript sections** – I included a section called Availability and requirements after the Conclusions.

**References** - I placed the references before the full stop.

**Figures** - I cropped the figure as closely as possible to minimize white space around the image.

Sincerely,

Peter L. Elkin, MD.
Professor of Medicine
Mayo Clinic College of Medicine